Author's response to reviews

Title: Serotypes and Antibiotic Resistance of Streptococcus pneumoniae Isolates from Patients at a General Hospital in Guangzhou, from 2011 to 2013

Authors:

Songyin Huang (hsongyin@126.com)
Xiaoqiang Liu (liu_1507@163.com)
Weisi Lao (laoweisi@21cn.com)
Suhua Zeng (zeng1507@163.com)
Huiqi Liang (846833194@qq.com)
Rihui Zhong (13570217766@139.com)
Xinlu Dai (443848397@qq.com)
Xiquan Wu (xiquan_wu@126.com)
Hongyu Li (lehyu@126.com)
Yandan Yao (yyandan@126.com)

Version: 2
Date: 29 August 2014

Author's response to reviews: see over
Dear Editor,

Thank you for quickly responding (MS: 1357803864140674) and providing us with instructive suggestions. We agree with the comments from the editor, and have answered all the questions of the editor. Please receive the detailed point-by-point response to the editor’s comments as follows. Therefore, we submit a revised manuscript.

Point-by-point description of the changes made:

1. **Line Numbering:** Please revise your manuscript to include line and page numbers.

   We have revised our manuscript to include line and page numbers.

2. **In addition, please include all authors email addresses on your title page.**

   We have included all authors email addresses on our title page.

Sincerely,

Yandan Yao
General Surgery
Sun Yat-Sen Memorial Hospital
Sun-Yat-Sen University
107 Yanjiang West Rd.
Guangzhou, Guangdong 510120
China